Author's response to reviews

Title: Characteristics of severely obese children and adolescents in the Netherlands (HELIOS project)

Authors:

- Sabine Makkes (s.makkes@vu.nl)
- Carry M Renders (carry.renders@vu.nl)
- Judith E Bosmans (j.e.bosmans@vu.nl)
- Olga van der Baan-Slootweg (obaan@heideheuvel.merem.nl)
- Jacob C Seidell (j.c.seidell@vu.nl)

Version: 2 Date: 2 August 2012

Author's response to reviews: see over
Dear editor,

On behalf of myself and my co-authors, I resubmit our article with revision entitled “Characteristics of severely obese children and adolescents in the Netherlands (HELIOS project)” to your journal BMC Pediatrics.

I added the section ‘Authors’ contributions’ in the manuscript behind the ‘Competing interests’ section.

Sincerely,

Sabine Makkes

Sabine Makkes, MSc
Junior Researcher
Section Prevention and Public Health
Department of Health Sciences
Faculty of Earth and Life Sciences
VU University Amsterdam
De Boelelaan 1085
1081 HV Amsterdam
Room T-641
Telephone number: +31 20 598 3673
Fax number: +31 20 598 6940
Email: s.makkes@vu.nl